Author’s response to reviews

Title: A prospective, randomized comparison of the LMA-ProtectorTM and i-gelTM in paralyzed, anesthetized patients

Authors:

Jee-Eun Chang (mistyblue15@hanmail.net)
Hyerim Kim (gracesnail@naver.com)
Jung-Man Lee (yzsmulsik@naver.com)
Seong-Won Min (swmin@brm.co.kr)
Dongwook Won (noninvasiveane@gmail.com)
Kwanghoon Jun (3young@hanmail.net)
Jin-Young Hwang (mistyblue15@naver.com)

Version: 2 Date: 21 May 2019

Author’s response to reviews:

BANE-D-18-00494R1

A prospective, randomized comparison of the LMA-ProtectorTM and i-gelTM in paralyzed, anesthetized patients

Dear Dr. Guangde Tu,

We are very glad to hear from you, and we thank you for your consideration.

According to your comment, we revised the manuscript.

We thank you again for your interest and consideration.

Sincerely,

Hwang
Editor Comments:

1. Declarations: "Ethics approval and consent to participate", please also make clear if consent to participate was obtained from each patient.

=> We revised it as follows.

Ethics approval and consent to participate

The present study was approved by the Ethics Committee of SMG-SNU (Seoul Metropolitan Government Seoul National University) Boramae Medical Center in Seoul, Republic of Korea (no. 20170228/26-2017-33/032) on 24 Mar 2017. Written informed consent was obtained from all patients.

2. Declarations: "Authors’ contributions", please add a statement that "All authors read and approved the final manuscript."

=> We added it as follows.

Authors’ contributions

JE Chang designed the study, conducted the study, analyzed the data, and wrote and revised the manuscript. H Kim designed the .................... JY Hwang designed the study, conducted the study, analyzed the data, and wrote and revised the manuscript. All authors read and approved the final manuscript.

3. Title "Introduction" should be "Background"

=> We corrected it.

4. Please provide us the clean final version without tracked changes.

=> According to your advice, we submitted a clean final version.

Reviewer reports:
Jong-Yeop Kim (Reviewer 1): The authors of this paper made considerable revisions and modifications based on the reviewers' comments, making this article interesting and relevant to SGA users.

=> Thank you.